Author’s response to reviews

Title: COPD Online-rehabilitation versus Conventional COPD rehabilitation - rationale and design for a multicenter randomized controlled trial study protocol (CORe trial)

Authors:

Henrik Hansen (henrik.hansen.09@regionh.dk)
Therese Bieler (heresa.Bieler@regionh.dk)
Nina Beyer (ninabeyer.privat@gmail.com)
Nina Godtfreden (Nina.Skavlan.Godtfredsen@regionh.dk)
Thomas Kallemose (thomas.kallemose@regionh.dk)
Anne Frølich (Anne.Froelich@regionh.dk)

Version: 2 Date: 31 Aug 2017

Author’s response to reviews:

August 31st 2017

To BMC Pulmonary Medicine

Dear Editor

Thank you for your reviewed comments on our manuscript entitled “COPD Online-rehabilitation versus Conventional COPD rehabilitation – rationale and design for a multicenter randomized controlled trial study protocol (CORe trial)”. In this letter we have answered your editorial comments.

1. For author contributions can you please use author's initials instead of full names

Thank you, full authors names in the author contribution section are now replaced with initials in the revised manuscript.

2. Please attach Figure 1 as a separate file with the figure legend at the end of the body of the manuscript instead of underneath the figure

Figure 1 is removed from the revised manuscript and attached in a separate word document and the figure legend placed after the reference list in the manuscript.
We hope you find the requested revision satisfying and hereby submit the revised manuscript in a “clean” version to be considered for publication in BMC Pulmonary Medicine.

We are looking forward to hearing from you.

Yours sincerely

Henrik Hansen
Corresponding author
Research Unit for Chronic Diseases and Telemedicine, University Hospital Bispebjerg and Frederiksberg, Bispebjerg Bakke 23, 2450 Copenhagen NV, Denmark. Phone +4528946780. Fax +4535316997. E-mail: Henrik.hansen.09.regionh.dk